\documentstyle[12pt,twoside]{article}
\input{macros-course}

\begin{document}

\handout{student-info}{Information Sheet}{September 5, 1996}

In addition to using this information to determine section assignments,
we will distribute a class list with phone numbers and e-mail addresses 
to facilitate collaboration.
If you do not want your phone number or e-mail address distributed by us,
please say so below.


\bigskip

\def\lineitem#1{
\hbox to \hsize
  {\hbox{#1} \hrulefill }
\medskip}

\lineitem{Name}
\lineitem{Department and year}
\lineitem{Phone number}
\lineitem{E-mail address}

\bigskip
Please list all Course VI and Course XVIII subjects you have taken or 
are currently taking.  Circle those you are currently taking.

\vskip 1.5in

Each student must attend a one-hour tutorial on Monday.  
Please circle the hours you prefer and cross out those you cannot attend.

\vskip 0.2in
{\Large
 \qquad 1 \qquad 2 \qquad 3 }


\vskip .5in 

Recitations will be held on Fridays.  Recitations are scheduled for
1:00 (26-210), 2:00 (26-210 and 34-301) and
3:00 (34-301).  Please circle the hours you prefer and cross out those
you cannot attend.

\vskip 0.2in
{\Large
\qquad 1 \qquad 2 \qquad 3 }

\end{document}
